Author’s response to reviews

Title: Effect of health education on severe thalassemia prevention and control in communities in Cambodia

Authors:

Kimhaung Cheng (kim3ley@gmail.com)
Supan Fucharoen (supan@kku.ac.th)
Kanokwan Sanchaisuriya (kanokwan@kku.ac.th)
Goonnapa Fucharoen (goonnapa@kku.ac.th)
Pattara Sanchaisuriya (pattara@kku.ac.th)
Arunee Jetsrisuparb (arujet@kku.ac.th)

Version: 3 Date: 17 Oct 2017

Author’s response to reviews:

Dear Editor,

I am very pleased to hear that you have very kind considerations accepting our article: AOPH-D-17-00094R2- Effect of health education on severe thalassemia prevention and control in communities in Phnom Penh, Cambodia with a minor correction. Please accept my sincere thanks and appreciation for your generosity.

I have deleted Phnom Penh from the title as per your suggestion already.

The important thing I would like to ask you to send me very soon is the acceptance letter for this manuscript which is important for Kimhaung Cheng to demonstrate it for her graduation. Concerning the page processing please let me know how to pay. I will go abroad between October 19 and 27, therefore I can pay before or after this period.

Wish you all the best,

Arunee Jetsrisuparb